Author's response to reviews

Title: Is seeing a specialist nurse associated with positive experiences of care? The role and value of specialist nurses in prostate cancer care

Authors:

Carolyn Tarrant (ccp3@le.ac.uk)
Paul Sinfield (pks4@le.ac.uk)
Shona Agarwal (sa144@le.ac.uk)
Richard Baker (rb14@le.ac.uk)

Version: 3 Date: 6 March 2008

Author's response to reviews:

Dear Dr Bucceri

We are very pleased that you have accepted our paper for publication. We have addressed the outstanding point raised by reviewer 1 in our discussion section, by noting the possibility of selection bias resulting from the 60% response rate. We have also checked the formatting of the manuscript. We hope it now conforms to the journal requirements but please let us know if there is anything we have missed.

Thnks and best wishes

Carolyn Tarrant
(on behalf of the authors)